

Citation: Ajina S, Bridge H (2018) Blindsight relies on a functional connection between hMT+ and the lateral geniculate nucleus, not the pulvinar. PLoS Biol 16(7): e2005769. https://doi.org/10.1371/ journal.pbio.2005769

Academic Editor: Christopher Pack, McGill University, Canada

Received: February 20, 2018

Accepted: July 9, 2018

Published: July 25, 2018

Copyright: © 2018 Ajina, Bridge. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: Wellcome Trust https://wellcome.ac.uk/ (grant number 096001/Z/11/Z). to SA. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. Medical Research Council https://mrc. ukri.org/ (grant number MR/K014382/1). to HB. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. Academy of Medical RESEARCH ARTICLE

Blindsight relies on a functional connection between hMT+ and the lateral geniculate nucleus, not the pulvinar

Sara Ajina*, Holly Bridge

Wellcome Centre for Integrative Neuroimaging, FMRIB, Nuffield Department of Clinical Neurosciences, University of Oxford, Oxford, United Kingdom

* sara.ajina@ndcn.ox.ac.uk

Abstract

When the primary visual cortex (V1) is damaged, the principal visual pathway is lost, causing a loss of vision in the opposite visual field. While conscious vision is impaired, patients can still respond to certain images; this is known as 'blindsight'. Recently, a direct anatomical connection between the lateral geniculate nucleus (LGN) and human motion area hMT+ has been implicated in blindsight. However, a functional connection between these structures has not been demonstrated. We quantified functional MRI responses to motion in 14 patients with unilateral V1 damage (with and without blindsight). Patients with blindsight showed significant activity and a preserved sensitivity to speed in motion area hMT+, which was absent in patients without blindsight. We then compared functional connectivity between motion area hMT+ and a number of structures implicated in blindsight, including the ventral pulvinar. Only patients with blindsight showed an intact functional connection with the LGN but not the other structures, supporting a specific functional role for the LGN in blindsight.

Author summary

When the primary visual cortex (V1) is damaged in one hemisphere, we lose the ability to see one half of the world around us. Clinical tests show that in this blind region of vision, we cannot see even the brightest flashes of light. However, many years of research have shown that individuals who are blind in this way may still respond to certain images in the 'blind' area of vision, even though they are often unable to describe what they 'see' and may be unaware of seeing anything at all. This is called blindsight, and researchers are trying to understand the pathways underlying this phenomenon. A recent study mapped a physical pathway of connections in the brain that could account for blindsight in humans. However, the functional nature of this pathway has never been shown. In this study, we assess a group of patients with damage to V1, some of whom demonstrate blindsight and some of whom do not. We compare neural responses and functional connectivity and show that a functional connection in this pathway is critical for blindsight. We also reveal new insights into how speed and motion are likely to be processed in the healthy brain.



Sciences https://acmedsci.ac.uk/ (grant number). to SA. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. Royal Society https://royalsociety.org/ (grant number). HB was a Royal Society University Research Fellow. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. NIHR Oxford Biomedical Research Centre (grant number). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: 2AFC, two-alternate forced choice; BET, Brain Extraction Tool; BOLD, blood oxygen level–dependent; COPE, contrast of parameter estimate; fMRI, functional MRI; FMRIB, Functional Magnetic Resonance Imaging Centre of the Brain; FSL, FMRIB's Software Library; LGN, lateral geniculate nucleus; MNI, Montreal Neurological Institute; MT, middle temporal area; ROI, region of interest; SC, superior colliculus; SEM, standard error of the mean; V1, primary visual cortex; VFI, Visual Field Index.

Introduction

Damage to the primary visual cortex (V1) that may occur following a stroke causes visual loss in the corresponding part of the visual field (homonymous hemianopia, [1]). However, extensive research has shown that some patients retain an ability to respond to images inside their scotoma, even though they may not consciously see them [2]. This phenomenon is called blindsight, and recent work applied diffusion MRI and tractography in patients with V1 damage to try to uncover which pathways may underlie this residual visual function [3]. A connection between the lateral geniculate nucleus (LGN) and human motion area, hMT+, was found to be intact in patients with blindsight but was absent or impaired in patients without blindsight. The other pathways tested, which included a connection between hMT+ and the superior colliculus (SC), and with hMT+ in the opposite hemisphere, did not show this pattern. Unfortunately, a limitation of diffusion MRI is that it investigates purely structural connections, which may not relate directly to the function under investigation [4]. Furthermore, seed-based tractography is restricted to pathways chosen by investigators (see also [5-7]), which in this case did not include a connection with the thalamic pulvinar nucleus. Neither a specific role for the LGN nor a functional connection to hMT+ has been shown in human blindsight and would significantly advance our understanding of how patients respond to visual images in the absence of V1.

The current study investigated behavioural and functional MRI responses to speed of motion in a group of patients with V1 damage in adulthood (n = 14), and healthy age-matched controls (n = 8). Patients were categorised as blindsight positive or negative according to their ability to detect the visual stimulus within their blind visual field. We then compared measures of activity and functional connectivity between the two patient groups and healthy controls.

Patients with blindsight demonstrated significant fMRI activity in hMT+ in the damaged hemisphere, with a relatively preserved hMT+ response to speed in the blind hemifield. Critically, patients with blindsight also showed intact functional connectivity between hMT+ and LGN in the damaged hemisphere, which was absent in patients without blindsight. This was specific to the LGN, as both patient groups demonstrated preserved functional connectivity between hMT+ and (i) ventral pulvinar, (ii) SC, and (iii) contralateral hMT+, which was no different than in healthy controls. The pulvinar, in particular, is implicated in human and nonhuman primate studies in which V1 is damaged early in life [8,9]; however, this region is yet to be evaluated in adult-onset blindsight. Our findings support a critical functional role for the LGN and its specific connection with hMT+ in adult human blindsight, reinforced by recent evidence for an intact anatomical connection between these structures [3].

Results

Behavioural performance for stimuli in the blind hemifield was measured using two two-alternate forced choice (2AFC) paradigms (S1 Fig). Participants viewed a central fixation cross whilst being shown an aperture of black dots on a grey background, moving at speeds of 0, 4, 8, 20, or 32°/s. Patients were categorised as blindsight positive or negative according to their ability to detect which of two intervals contained the moving dots (see 'Materials and methods' for more details). According to this measure, 8 out of 14 patients with unilateral V1 damage and homonymous hemianopia were labelled as 'blindsight positive'. Optimum performance for both tasks (i.e., most significant) was at intermediate speeds of 8 and 20°/s, and performance tended towards a quadratic relationship with speed (F = 2.6, *p* = 0.09, df = 2, <u>S1C and</u> <u>S1D Fig</u>). We performed functional MRI (fMRI) using the same visual stimuli in all 14 patients, with images presented separately to the blind or sighted hemifield in order to relate neural activity and functional connectivity to blindsight function.

Residual responses to motion after V1 damage

Blindsight-positive patients demonstrated significant fMRI activity in contralateral hMT+ for moving versus static dots in the blind hemifield (Fig 1Ai and 1Aii). Overall hMT+ blood oxygen level-dependent (BOLD) signal change for all 5 speeds was significantly greater than baseline (Fig 1Aiii, t = 3.0, p = 0.02, df = 7). Activity was somewhat lower in intensity and spatial extent than the sighted field or healthy controls (Fig 1C), but there was no significant difference between blind and sighted responses in blindsight-positive patients (Fig 1Aiii t = 1.3, p = 0.2, df = 39). Whilst the absence of a significant difference may result from a lack of power (n = 8), blindsight-negative patients with a lower 'n' (Fig 1B) showed a marked effect of hemifield (paired t = 2.9, p = 0.007, df = 29) and no demonstrable hMT+ activity for moving versus static dots in the blind hemifield (Fig 1Bi and 1Bii) nor averaged across all conditions (Fig 1Biii, t = 0.6, p = 0.6, df = 5).

In addition to hMT+ activity, blindsight-positive patients with right V1 damage (Fig 1Aii, n = 2) showed activity in Jülich-defined right V2 and left V4, although there was no such



Fig 1. fMRI responses to motion for patients with V1 lesions and controls. Results are shown separately for (**A**) blindsight-positive patients, (**B**) blindsight-negative patients, and (**C**) healthy controls. (**i**) Significant activity for moving versus static dots in the blind right hemifield of patients with left V1 lesions and (**ii**) the blind left hemifield of patients with right V1 lesions. Mixed effects analyses, P < 0.001 uncorrected for a priori regions of interest, elsewhere cluster-corrected p < 0.01. Shaded blue areas are binarized Jülich-defined probabilistic maps of hMT+, radiological convention. (**iii**) Mean contralateral hMT+ signal change averaged across all five stimulus conditions, comparing sighted (blue) and blind (red) hemifields ± SEM. In controls, 'Left HF' refers to left hemifield, 'Right HF' is right hemifield. * significant activity above baseline p < 0.05, $\psi p \le 0.001$, ns = not significant. *P* values are from *t* tests. (**iv**) Mean signal change in contralateral hMT+ as a function of stimulus speed, shown separately for each hemifield (blue squares are sighted hemifield, red diamonds are blind hemifield). Responses to static stimuli are dotted (left/sighted hemifield) or dashed (right/blind hemifield) lines. Underlying data for iii and iv can be found in <u>S1 Data</u>. fMRI, functional MRI; SEM, standard error of the mean; V1, primary visual cortex.

https://doi.org/10.1371/journal.pbio.2005769.g001

activity in patients with left V1 damage (n = 6). In blindsight-negative patients, visual-evoked responses also appeared within small regions of ipsilesional V4 and V2, as well as the occipital pole and inferior parietal lobule, implying that activity in these regions was insufficient for motion perception.

It was noteworthy that overall hMT+ signal change in the undamaged hemisphere was also slightly reduced in both patient groups compared to age-matched controls. A possible explanation is that unilateral V1 damage can negatively impact sighted processing in the opposite hemisphere, perhaps via a disturbance of interhemispheric interactions [10,11].

Speed of motion $(0^{\circ}-32^{\circ}/s)$ had a significant impact on hMT+ activity in the sighted hemifield of patients and equivalent left hemifield of controls (2-way ANOVA: F(4, 14) = 5.1, p < 0.001), with no effect of participant group (Fig 1iv, F(2, 14) = 2.2, p = 0.12). In the blind hemifield of patients (right hemifield of controls) there was a similar effect of speed (F(4, 14) = 3.5, p = 0.01) but also group (F(2, 14) = 7.4, p = 0.001). The pattern of responses in blindsightpositive and -negative patients differed markedly. Blindsight-positive patients showed a positive relationship between hMT+ signal change and speed (r = 0.8, 5 speeds), whilst the correlation coefficient was negative in blindsight-negative patients (r = -0.6). This difference was not simply driven by a difference between motion and static responses, as excluding the static conditions increased the significance even further (r = 0.67 versus r = -0.99, z = -2.44, p = 0.01). Specifically, blindsight-positive patients showed a relatively 'normal' hMT+ relationship with speed in the blind hemifield that was similar to the pattern in healthy controls (Pearson r = 0.89, p = 0.04, 5 speeds) and patients' own sighted hemifield (r = 0.98, p < 0.01). This was not the case for blindsight-negative patients, either when compared to controls (r = 0.21, p = 0.7) or to their own sighted hemifield (r = 0.29, p = 0.6).

Critical functional geniculo-extrastriate connectivity

To determine how activity in hMT+ correlated with subcortical activity, we examined the fMRI time series after stimulus-evoked responses had been regressed out. Specifically, we compared the residual pattern of activity in hMT+ with (i) LGN, (ii) ventral pulvinar, (iii) SC, and (iv) hMT+ in the opposite (undamaged) hemisphere, using subject-specific regions of interest (ROIs) (S3 Fig). We also performed whole-brain analyses to measure the voxels where neural activity most closely matched the time series of hMT+ and these subcortical structures.

LGN–V1 correlation in the nondamaged hemisphere (Fig 2A) was similar across all participant groups (F = 1.0, p = 0.4, df = 2). Bilateral hMT+ correlation was also very similar across all groups, indicative of preserved functional connection between hMT+ in patients irrespective of blindsight status (Fig 2B, F = 1.9, p = 0.2, df = 2).

There was a significant effect of participant group on LGN–hMT+ correlation (Fig 3A, F = 10.1, p = 0.001, df = 2). Pairwise analysis for the damaged hemisphere showed that blind-sight-negative patients had a significantly lower mean correlation coefficient compared to blindsight-positive patients (-0.03 ± 0.08 SE versus 0.25 ± 0.03 SE, t = 3.6, p = 0.003, df = 12) and remained at zero. In contrast, functional connectivity between hMT+ and ventral pulvinar (Fig 3B) or hMT+ and SC (Fig 3C) showed no effect of group (F = 0.5, p = 0.6, df = 2, for both ROIs) and no difference between blindsight-positive and -negative patients in the damaged hemisphere (pulvinar: t = 0.9, p = 0.4, SC: t = 0.5, p = 0.6, df = 12). Blindsight-negative patients also showed a hemispheric difference for LGN (t = 3.0, p = 0.01, df = 10) but not ventral pulvinar connectivity (t = 2.1, p = 0.06, df = 10).

This suggests that the key difference in functional connectivity between blindsight-positive and -negative patients was the presence of a functional connection between hMT+ and LGN in the damaged hemisphere. We performed an additional analysis without regressing out



Fig 2. Functional connectivity of the visual pathways in patients and controls. (A) Correlation of LGN and V1 in the same (undamaged) hemisphere over the entire fMRI timeseries, after stimulus-evoked activity has been regressed out. (B) Correlation of hMT+ bilaterally. Box plots show Fischer-corrected mean correlation coefficients comparing participant group ± SEM. Statistical symbols represent significance levels for one sample t-tests against baseline (zero): $\varepsilon p \le 0.0001$, * p < 0.001. Scatterplots are individual examples of fMRI signal in ROI1 versus ROI2. Each point represents a single fMRI volume. Plots for patients in **panel A** are correlations in the contralesional hemisphere. BS+ is blindsight positive, and BS- blindsight negative. Underlying data can be found in S2 Data. fMRI, functional MRI; LGN, lateral geniculate nucleus; ROI, region of interest; SEM, standard error of the mean; V1, primary visual cortex.

https://doi.org/10.1371/journal.pbio.2005769.g002

stimulus-evoked responses and found the same results (one-way ANOVA F = 4.3, p = 0.02, df = 2; paired t = 3.4, p = 0.005, df = 12), implying that in patients with blindsight this functional connection was also present during visual function. These analyses, however, have not evaluated whether differences in connectivity were specific to our predefined regions of interest or if they reflect a global process independent of the ROIs and our hypothesis. To address this, we performed an additional whole-brain mixed effects analysis measuring the voxels where neural activity most closely matched the time series of LGN, ventral pulvinar, and hMT+. This technique was



Fig 3. Functional connectivity of subcortical hMT+ pathways in patients and controls. (A) Correlation of LGN and hMT+. **(B)** Correlation of ventral pulvinar and hMT+. **(C)** Correlation of SC and hMT+. Box plots show Fischer-corrected mean correlation coefficients comparing each participant group ± SEM. Results are shown separately for the intact (solid boxes) and damaged hemispheres (striped boxes), and for left (striped) and right (solid) hemispheres in controls. Underlying data can be found in <u>S3 Data</u>. LGN, lateral geniculate nucleus; SC, superior colliculus; SEM, standard error of the mean.

https://doi.org/10.1371/journal.pbio.2005769.g003

used to generate seed region correlation maps, in which the 'seed' ROI should necessarily demonstrate a high correlation coefficient represented by a beta of 1 [12]. Co-active regions would similarly possess a high beta, with maps retaining a high spatial resolution since every voxel is tested [13]. As expected, group correlation maps showed a high beta in the 'seed' regions, reflecting consistency between subject-specific ROIs and their precise transformation to standard space (Fig 4). When LGN was the 'seed' region, control participants also demonstrated a relatively high beta in a small region of the calcarine cortex corresponding to retinotopically active V1 (Fig 4Aiv and



Fig 4. Seed region correlation maps for LGN and ventral pulvinar, in patients and controls. 'Seed regions' are (**A**) LGN in the damaged hemisphere (left in controls), (**B**) ventral pulvinar in the damaged hemisphere (left in controls), (**C**) LGN in the undamaged hemisphere (right in controls), (**D**) ventral pulvinar in the undamaged hemisphere (right in controls), (**D**) ventral pulvinar in the undamaged hemisphere (right in controls), (**D**) ventral pulvinar in the undamaged hemisphere (right in controls), (**D**) ventral pulvinar in the undamaged hemisphere (right in controls). Results are shown separately for controls (left column), blindsight-positive patients (middle column), and blindsight-negative patients (right column). Mixed effects analyses, displayed on average high-resolution structural scans transformed to MNI space (radiological convention). Shaded blue areas are binarized Jülich-defined probabilistic maps of hMT+. LGN, lateral geniculate nucleus; MNI, Montreal Neurological Institute.

https://doi.org/10.1371/journal.pbio.2005769.g004

4Civ), reflecting the functional geniculostriate pathway. A similar region of calcarine cortex was co-activated in the undamaged hemisphere of patients when the seed was LGN in the same hemisphere (Fig 4Cv and 4Cvi). For LGN in the damaged hemisphere of both patient groups, there was no demonstrable V1 co-activation (Fig 4Av and 4Avi), likely reflecting the damage to that region (and/or its input). In contrast, blindsight-positive patients showed a relatively high beta in hMT+ of the damaged hemisphere compared to other participant groups and a small region of calcarine cortex in the undamaged side (Fig 4Aii).

In contrast to the LGN, when ventral pulvinar was used as the 'seed', all groups showed robust co-activity in the SC, an area known to share an important connection with the pulvinar [14,15]. There was also notable connectivity with V1 in the undamaged hemisphere (Fig 4D, [16]) but no major connectivity with hMT+ (Fig 4B).

When hMT+ was used as the 'seed region' (Fig 5), controls demonstrated marked functional connectivity throughout the visual cortex, including hMT+ in the opposite hemisphere and V1 in both hemispheres, consistent with previous reports [17]. In subcortical regions, connectivity was also demonstrable in the LGN bilaterally, albeit to a lesser extent (Fig 5A and 5B, left column). No equivalent connectivity was seen in the ventral pulvinar or SC, although these regions were co-activated in both hemispheres of all three participant groups if a slightly lower beta threshold of 0.32 was used (rather than 0.35). This pattern of connectivity was likely to reflect the major visual pathway and its rich network of intra- and interhemispheric connections. A very similar pattern was demonstrated in blindsight-positive patients, except for a relatively low beta in calcarine cortex of the affected hemisphere, reflecting the region of tissue damage. This was less apparent in blindsight-negative patients. Blindsight-negative patients also showed relatively poor connectivity with LGN, particularly in the damaged hemisphere, where ipsilateral geniculate co-activity was not demonstrable even when a low threshold was applied (beta > 0.2).

Behavioural-neuroimaging comparisons

Of the eight blindsight-positive patients, only half could discriminate direction of motion above chance. We were interested in whether this subgroup showed greater BOLD activity and/or connectivity compared to blindsight-positive patients who were unable to discriminate motion direction. Indeed, these four patients did show slightly stronger motion responses in hMT+ ($0.3 \pm SE \ 0.13$ versus 0.1 ± 0.07 , t = 1.6, p = 0.2, df = 6), as well as slightly greater LGN/ hMT+ functional connectivity (0.27 ± 0.05 versus $0.22 \pm SE \ 0.06$, t = 0.6, p = 0.5, df = 6), although the differences were relatively small and nonsignificant.

To investigate the behavioural–neuroimaging association further, we performed a correlation analysis between mean behavioural performance in both experiments and fMRI activity and connectivity across all patients (n = 14). hMT+ motion responses showed a weak but positive correlation with behavioural performance (r = 0.38, p = 0.18), as did LGN–hMT+ connectivity (r = 0.38, p = 0.18). There was also a positive but weaker correlation between behavioural performance and pulvinar–hMT+ (r = 0.29, p = 0.31) or SC–hMT+ connectivity (r = 0.21, p = 0.47). More notable was a significant correlation between behavioural performance and the ratio of blind to sighted fMRI responses in hMT+ (r = 0.62, p = 0.018). In other words, patients with the strongest hMT+ activity for motion in the blind field relative to their sighted field performed best at behavioural assessments of blindsight using the same visual stimuli.

Relating lesion size to behavioural performance and functional connectivity

In order to determine the extent to which blindsight performance and the underlying neural mechanisms relate to the size of the lesions, we quantified the damage in each patient. Lesion



Fig 5. Seed region correlation maps for human motion area (hMT+), in patients and controls. 'Seed region' is **(A)** hMT+ in the damaged hemisphere (left in controls) and **(B)** hMT+ in the undamaged hemisphere (right in controls). Results shown separately for controls (left column), blindsight-positive patients (middle column), and blindsight-negative patients (right column). Upper rows show axial slices through early visual cortex and hMT+; lower rows show coronal slices through LGN. Mixed effects analyses, displayed on average high-resolution structural scans, transformed to MNI space (radiological convention). Shaded green areas are binarized Jülich-defined probabilistic maps of LGN. LGN, lateral geniculate nucleus; MNI, Montreal Neurological Institute.

https://doi.org/10.1371/journal.pbio.2005769.g005

size did not differ significantly between blindsight-positive and -negative groups (t = 1.5; df = 12; p = 0.16), although blindsight patients on average had smaller lesions (13,693 mm³ ± 1,825 mm³ SEM) than those without blindsight (21,212 mm³ ± 5355 mm³ SEM). This is illustrated by the summed lesion masks for each patient group on a standard space template (S2D and S2E Fig) and the individual lesion maps in structural space (S4 Fig). Reflecting the small difference in lesion size between the two patient groups, there was a moderate inverse relationship between lesion size and behavioural performance across both tasks (r = -0.48; p = 0.08). However, there was no relationship between lesion size and functional LGN–hMT+ connectivity (r = 0.21, p = 0.5), or hMT+ signal change (r = -0.24, p = 0.4). Thus, lesion size did not appear to be a critical factor in determining neural response.

Aside from lesion size, the degree to which lesions involve hMT+ and its innervating connections would also be critically important. Lesion maps (S4 Fig) suggest that the lesion of one blindsight-negative patient (P12) did encroach upon hMT+, while it appeared intact in the other five blindsight-negative patients, including its afferent white matter. P1 also had a large lesion extending to the posteromedial border of hMT+; however, this should not have impacted upon the subcortical or interhemispheric white matter connections (see S4 Fig for map). Accordingly, we quantified lesion involvement of hMT+ and its surrounding white matter in each patient (S5 Fig). We found no difference when comparing blindsight-positive and -negative patients (t = 1.7, p = 0.11, df = 12), and hMT+ lesion size showed no association with LGN-hMT+ functional connectivity (t = 1.6, p = 0.13, df = 13), hMT+ activity (t = 1.4, p = 0.19, df = 13), or behavioural performance (t = 1.6, p = 0.13, df = 13) in paired t tests using lesion size as the dependent variable.

When recruiting a relatively large group of patients, it is challenging to ensure a completely homogenous lesion pattern. However, any specific differences in lesion size or extent can be informative. P12, whose lesion may have encroached on hMT+, demonstrated weak functional connectivity (outside one SD of the mean) in all three subcortical pathways (pulvinar–hMT+: r = 0.004, SC–hMT+: r = -0.13) and between hMT+ bilaterally (r = 0.33), and this contributed to slightly lower averages for pulvinar, collicular, and interhemispheric (but not LGN) connectivity in blindsight-negative patients (see Fig 2B and Fig 3B and 3C). The variability amongst naturally occurring human V1 lesions has been highlighted as a limitation of human research compared to nonhuman primates [18]. It certainly emphasizes the limitations in carrying out individual case studies, which have predominated in the blindsight literature over the last several decades. However, the heterogeneity in the precise location of structural damage can also be extremely useful and has permitted patients to be classified according to their distinct residual visual performance. By determining which connections and characteristic fMRI responses are consistent amongst blindsight-positive and -negative patients, it may be possible to identify which underlying structures and pathways are involved.

To summarise our results, blindsight-positive patients showed (i) significant neural activity in hMT+ to motion stimuli in the 'blind' visual field, (ii) a relatively preserved response to speed in hMT+, and (iii) a correlation between resting BOLD signal in LGN and hMT+ in the damaged hemisphere. None of these findings were demonstrable in patients without blindsight. However, blindsight-negative patients did show functional connectivity between hMT+ and (i) SC and (ii) ventral pulvinar that was no different to healthy controls or patients with blindsight, suggesting that the LGN has a specific functional role in blindsight.

Discussion

This study tested a group of patients, all of whom sustained unilateral damage to V1 in adulthood. Critically, not all the patients demonstrated the same residual visual function, allowing us to relate brain activity to blindsight performance using the same visual stimuli. At the most basic level, we have shown that human motion area hMT+ does not require V1 to demonstrate a normal fMRI response to speed. More importantly, we have found additional evidence to support a specific functional role for the LGN in blindsight and to suggest that functional connectivity with the ventral pulvinar and SC is not sufficient for blindsight.

Intrinsic functional connectivity corresponds to a number of anatomical markers of connectivity, including anatomical traced pathways [19], axonal pathway density [20], and intracortical myelin content [21]. There is also evidence that the strength of correlation between functionally coupled regions relates to measures of cognition and behaviour and can be useful markers of brain system integrity in neurological and psychiatric disorders (e.g., see [22] for review). When two regions, such as left and right motor cortices, share a high correlation coefficient (r = 0.8, [23]) similar to left and right hMT+ in our study, this is interpreted as an example of strong functional interconnectivity. In contrast, a correlation coefficient of zero between LGN and hMT+ in blindsight-negative patients is similar to the correlation between motor and visual cortices in healthy individuals and implies a weak or absent functional connection [23]. The relationship between functional and structural connectivity measures is important to consider, as these can give different results. It is possible for normal-appearing functional connectivity to occur even when there is a known disruption in the physical connection [24], perhaps explained by intact polysynaptic anatomic connections or common feed-forward projections [25]. In previous work using diffusion MRI, we could only identify streamlines between SC and hMT+ in 7 out of 9 healthy controls and 14 out of 17 patients (in the intact hemisphere) [3]. In the current study, we observed intact functional connectivity for this pathway in all participants. This is an important discrepancy, as we cannot be certain which methodology reflects the underlying physiology most accurately. It may be possible for a functional connection to exist between SC and hMT+ via an indirect pathway while direct, diffusion MRI-traceable connections were absent. In support of this, the connection between SC and hMT+ is believed to relay via a synapse in the pulvinar nucleus [26]. Another explanation for the lack of consistency in diffusion tractography was that the size of the tract rendered it difficult to trace, and therefore, differences in brain geometry had a greater effect on identification. These points emphasise the importance of functional connectivity as a physiological measure of connected brain regions and the limitation of using diffusion MRI tractography alone.

It is well described that hMT+ remains active after V1 damage in blindsight [27–31]. More recently, we have started to explore how hMT+ responds to common stimulus parameters after V1 is damaged [32,33]. We have previously reported that normal hMT+ responses to global motion [32] and contrast sensitivity [33] are dependent on V1, as both become abnormal after unilateral V1 damage, even in the presence of blindsight [33]. The current study, however, has shown that a classical hMT+ response to speed remains intact in patients who demonstrate blindsight. This is consistent with the recent finding in healthy individuals that visual motion information from LGN can reach hMT+, bypassing V1, in response to both slow and fast speeds of motion [34]. There are also two reports in the literature that have measured macaque middle temporal area (MT) neuronal responses to speed after striate cortex removal, and both imply that speed tuning may be retained, including responses to slow motion [35,36]. Together, these results suggest that visual input from regions other than V1 are sufficient for normal patterns of response to speed, but only in patients demonstrating blindsight.

The LGN is often implicated in blindsight as a possible nonstriate source of input to the extrastriate cortex [3,37,38]. Neuroanatomical studies suggest that there is a direct connection between the LGN and extrastriate cortex [39,40], including specific projections to middle temporal area (MT) [41-43]. However, its relatively small size and deep location makes it particularly challenging to image [44]. Our results support a functional role for the LGN in blindsight for several reasons. Firstly, blindsight-positive patients showed retained functional connectivity between LGN and hMT+, whilst this was absent in blindsight-negative patients. This was demonstrated both at a whole-brain level and using an ROI analysis. Secondly, these results were specific to the LGN, as blindsight-negative patients showed retained functional connectivity with both ventral pulvinar and SC that was no different to healthy controls or patients with blindsight, despite both structures being implicated in blindsight [5,8]. We are confident that we were able to distinguish the LGN from the ventral pulvinar since the subcortical regions co-activating with hMT+ showed a strong correspondence to Jülich probabilistic maps of the LGN using whole brain analyses (Fig 5Av) [45]. Similarly, seeding in the pulvinar (but not the LGN) showed robust co-activation with SC throughout all participant groups.

One limitation was that directionality of the functional connections could not be determined. Theoretically, hMT+ in blindsight could be driven by non-geniculate input but show intact functional connectivity with LGN due to retained feedback [46]. On balance, this seems unlikely, as in blindsight-negative patients, the LGN was the only subcortical region to lose a functional connection with hMT+. Connectivity with the undamaged hemisphere was also no different to controls, consistent with our recent report of a patient with bilateral V1 damage [47]. In any case, it is suggested that feedback projections from MT to LGN may require V1 [46]. A further challenge in this study was the relatively noisy subcortical signal, which was susceptible to motion artefact. This was, however, relevant to all subcortical regions, and our measures of functional connectivity proved robust and consistent between individuals.

The role of the pulvinar in residual vision and blindsight has been emphasized after identifying a direct connection with MT in monkeys [24,48,49], particularly since the connection between the medial division of the inferior pulvinar and MT is enhanced following early lesions of V1 [8]. The current study used a ventral pulvinar mask consistent with the literature [50]. We also employed whole-brain analyses to determine which structures showed functional correlation with hMT+. The role of the pulvinar therefore requires further investigation, as it may not carry sufficient information to facilitate significant behavioural performance in the tasks used here. However, it was interesting that blindsight-negative patients showed weak hMT+ activity to slow motion (4°/s, Fig 1Biv). Perhaps such an abnormal speed response could be supported by a functional connection with the pulvinar and/or SC, which appeared to be intact in the blindsight negative group. Macaque MT neurons show a variety of speedtuning curves that include the pattern seen in blindsight-negative patients (with peak responses at 4°/s) and more classical response with peaks at 8 or 20°/s (e.g., [51,52]). The underlying basis for these distinct neuronal subsets is unknown. One possibility is that they reflect different inputs, e.g., from intact V1, or direct subcortical nuclei such as LGN, SC, and pulvinar. V1 normally exhibits a similar preference for slow speed, with a negative slope similar to hMT+ in blindsight-negative patients [53]. Inferior pulvinar neurons also possess a range of speed-sensitivities, mostly preferring slower motion under 32°/s [54,55], not dissimilar to V1 responses. It is suggested that those responses may arise from striate cortex inputs [56], and it was interesting that blindsight-negative patients showed evidence of peripheral V1 activation, making this a possible source of input. Macaque LGN neurons also demonstrate a range of speed preferences, particularly in magnocellular neurons, but most prefer somewhat faster motion [57]. The observation that blindsight-positive and -negative patients showed a different speed-tuning curve in hMT+ despite equivalent preservation of hMT+ and its surrounding white matter suggests that speed-tuning is unlikely to arise de novo in hMT+ but is perhaps determined by the innervating pathway and its subcortical structures.

It is also possible that early and late V1 lesions result in different pathways carrying residual visual function. All the patients in this study acquired V1 lesions in adulthood. It would be useful to compare to patients with congenital or early life lesions, such as the recent case of a child with extensive damage to the occipital lobe bilaterally who showed preserved visual function [9]. This child showed a reduction in streamlines between LGN and hMT+ measured with diffusion imaging but showed an increase between the medial portion of the inferior colliculus and hMT+ in one hemisphere.

A small number of studies have implicated the SC in 'non-aware' or indirect blindsight [5,58,59]. However, those patients sustained brain injury at birth or in early childhood. After striate cortex removal in the macaque, directionally selective responses in MT stop if the SC is additionally destroyed [35,60], while there is no effect resulting from the removal of the SC alone. Behaviourally, monkeys who retain saccadic eye movements towards a target in their blind field after a V1 lesion lose this ability and the potential for recovery if the ipsilesional SC is also inactivated [61,62]. These findings could be reconciled with studies implicating the LGN in blindsight if the SC provides input to LGN in a collicular–geniculate–extrastriate

pathway [37]. Indeed, Kato and colleagues [62] suggest two complementary pathways, one collicular–geniculate-hMT+ and the other collicular–pulvinar–hMT+, which could be consistent with our measures of functional connectivity between these structures and the distinct hMT + response pattern in blindsight-positive and -negative patients.

Lastly, behavioural sensitivity to speed showed an 'inverted-U' pattern that was consistent with previous reports [63]. This was similar to characteristic speed responses measured in hMT+ using fMRI [53,64]. Patients also demonstrated a linear relationship between blindsight performance and relative hMT+ activity, which extends previous reports simply categorising patients with or without functional activity and significant accuracy [65]. This has important implications for rehabilitation, as it may be that enhancing this residual activity and functional connection could improve detection and discrimination performance.

Conclusions

In summary, we identified a functional connection between LGN and hMT+ in patients with blindsight that was absent in patients without blindsight, despite a retained functional connection with ventral pulvinar and SC. This supports a critical functional role for the LGN in human blindsight, and in particular its connection with hMT+, reinforced by recent evidence for an intact anatomical connection between these structures [3]. Our results also revealed that hMT+ does not require intact V1 for a normal speed response, although it does require a functional connection with the LGN. This suggests that the LGN may support motion-selective input to hMT+ in the absence of V1. These results focus on behavioural and neural responses to visual motion, which is a critical component of blindsight (see [66] for recent review). In future work, it will be necessary to explore how such pathways interact with other aspects of blindsight function and whether distinct tasks or stimuli might engage separate mechanisms in the absence of V1.

Materials and methods

Participant details

Fourteen patients with adult-onset unilateral V1 damage took part in this study (see S1 Table for details). The location of any additional non-V1 damage is shown in S2D and S2E Fig, and S4 Fig. No patients sustained damage to subcortical structures, including the LGN and pulvinar. Average age at the time of participation was 55.6 years \pm 15.2 SD; average time after pathology onset was 49 months (6–252 months). Eight age-matched, healthy participants (50.1 \pm 14.6 SD years) served as controls. Written informed consent was obtained from all participants, and ethical approval was provided by the Oxford Research Ethics Committee (Ref B08/H0605/156). All experiments adhered to the Declaration of Helsinki.

Stimuli and experimental design

Visual stimuli were generated using MATLAB (Mathworks) and the Psychophysics Toolbox [67,68]. Each dot was 0.075° in diameter and had an infinite lifetime, with an average dot density of 8 dots/°². Visual stimuli consisted of an aperture of 5° or 8° diameter containing static or coherently moving black dots (luminance 0.5 cd/m⁻²) at 4, 8, 20, or 32°/s on a uniform grey background of luminance 50 cd/m⁻². Stimuli were positioned inside a region of dense visual field loss in patients a minimum of 2.5° from fixation (S6 Fig). The extent to which stimuli covered the scotoma (as a percentage) was estimated for each patient from the Perimetry Visual Field Index (VFI), provided in S6 Fig. Stimulus size and position was matched as closely as possible in eight age-matched controls (S2A Fig), with no significant difference in distance between fixation and stimulus edge (x or y coordinates) when comparing patients to controls (mean x = 3.6 ± 0.30 SE patients versus 3.8 ± 0.4 SE controls, t = 0.4, p = 0.7, df = 20; mean y = 0.65 ± 0.66 SE patients versus 0.49 ± 0.68 SE controls, t = 0.2, p = 0.9, df = 20). Stimuli in blindsight-negative patients were also no deeper into the visual field than blindsight-positive patients (stimulus edge $4.0^{\circ} \pm$ SD 0.9 in blindsight-positive versus $3.5^{\circ} \pm$ SD 1.2 in blindsight-negative patients, t = 0.9, p = 0.4, df = 12).

To select the stimulus location in patients, we required a perimetry threshold p < 0.005 or < -20dB (which ever was more stringent) for pattern deviation compared to age-matched controls at the stimulus location. This meant that the patients in our study were unable to see even the brightest unattenuated stimuli at that location in the visual field. To verify that we had not inadvertently chosen locations in blindsight-positive patients that were more sensitive than those in blindsight-negative patients, we calculated the average pattern deviation by taking the value closest to the stimulated location using Humphrey Perimetry. The residual visual sensitivity was no different in the two groups (-32.8dB ± SE 0.8 blindsight positive versus -33.2dB ± SE 0.9 blindsight negative, t = 0.3, p = 0.8, df = 12).

Outside the scanner, two behavioural experiments were performed: (1) 2AFC temporal detection and (2) 2AFC direction discrimination (S1 Fig). The experiments were conducted on the same day as scanning, using a 60-Hz CRT monitor at a viewing distance of 68 cm. Throughout behavioural experiments, participants were asked to maintain fixation, with the investigator observing this in real time using an Eyelink 1000 Eye Tracker (SR Research Limited, Ontario, Canada). Anyone making even a small eye movement into their damaged hemi-field was given specific instruction not to do so, and it was explained that these data would have to be discarded.

At the start of the experiment, an identical, static test stimulus was used to confirm that patients were unable to see the stimulus at its selected size and location in the visual field. This was done using a predicted aperture size and locus based upon prior perimetry results. Stimulus location had to be restricted to the boundary of the fMRI display, which subtended 23° horizontally and 13° vertically. This influenced whether a 5° or 8° diameter stimulus was chosen, as the stimulus had to stay inside the 'blind' field while remaining on screen. The stimulus of choice was an 8° diameter aperture, but if this was not possible, the stimulus was reduced to 5° diameter. If the criteria were unachievable using either stimulus size, the patient was excluded from the study. If the patient was able to see any part of the test stimulus whilst fixating on the central cross, the aperture was repositioned 0.5° deeper into the scotoma (according to the Perimetry report) until the patient could no longer see any part of the stimulus at all. Any trials with eye position more than 1° from fixation were excluded from analysis.

Experiment 1: 2AFC temporal detection. Patients were asked to indicate whether a stimulus appeared in the first or second time interval using a two-alternate forced choice paradigm (S1A Fig). If they saw nothing, they were instructed to guess. Onset of each interval was indicated by a 500-ms auditory tone, with 300 Hz marking onset of the first interval and 1,200 Hz marking the onset of the second. Visual stimuli appeared for 500 ms with jittered onset while the participant fixated on a central black cross. The allocated interval (first or second) was generated at random. Stimulus speed was altered parametrically between the five conditions at random, with an average of 20 trials per condition.

Experiment 2: 2AFC direction discrimination. Patients were asked to indicate whether motion direction was horizontal or vertical (S1B Fig). Again, if they saw nothing, they were instructed to guess. Visual stimuli appeared for 500 ms with jittered onset whilst the participant fixated on a central black cross. Stimulus speed was altered parametrically between the four motion conditions at random, with an average of 20 trials per condition.

Blindsight definition

The presence or absence of residual visual function (blindsight) was determined according to patients' ability to detect stimuli above chance, i.e., Experiment 1. Specifically, this was defined as achieving either an average score or a score for individual conditions that was significantly above chance, using a statistical threshold of p < 0.05 and a cumulative binomial distribution. This was an identical method to our previous work, except that the stimulus was moving dots rather than a drifting Gabor [33].

We selected stimulus location based upon perimetry results, as detailed above. Necessarily, this meant that all patients showed the same abnormal visual performance for their test locations. Aside from demonstrating the same visual sensitivity on perimetry, stimuli in blind-sight-negative patients were no deeper into the visual field than they were for blindsight-positive patients, suggesting this was not a critical factor for the difference in behavioural performance.

Previous blindsight studies have employed a variety of visual stimuli (moving dots, gratings, moving bars, high luminance targets [31,37,63,69]) and a number of different techniques for assessment, including 2AFC, indirect behavioural performance, saccadic eye movements, and navigational performance (e.g., [5,70–72]). It is also common to target only one retinal location in blindsight testing [31,37,63,69,70]. The critical point for the definition of blindsight is that patients show significant performance despite absent visual capacity in the targeted region of the visual field. We ensured that a conservative threshold was used to target truly 'blind' regions of scotoma, which we demonstrated to be no different in patients with or without blindsight. Since we were particularly interested in the role of hMT+, we assessed whether a moving stimulus could be detected without awareness for our definition of blindsight.

Using these criteria, eight patients were categorized as 'blindsight positive', as they could detect the stimulus inside their blind hemifield significantly above chance (P2, P3, P5, P8, P10, P11, P13, P14). Of these individuals, four could also discriminate motion direction above chance (Experiment 2; P5, P8, P10, P14). With regard to subjective awareness, only two patients reported any awareness of the stimuli during the experiment (P3 and P10). Both had been completely unaware of static dots in the pre-experiment assessment, in which static dots were positioned at the same coordinates, without fast onset/offset. For moving stimuli, P10 reported knowing that something was there, but was unable to distinguish what it was. P3 also could not describe what she saw, suggesting she had been looking at 'streaks or shadows'. Both patients were at ceiling on the detection task, but P3 remained at chance on the direction discrimination task.

Of note, five of the patients (P3, P8, P10, P11, P13) took part in a previous fMRI study [33], in which they also demonstrated significant blindsight performance for detection of a drifting Gabor.

Behavioural eye-tracker results

Eye movements were defined as a change in fixation towards the scotoma of 1 degree or more. This would capture all eye movements irrespective of their type, i.e., saccadic, slow drift, nystagmus. The threshold of 1 degree ensured that stimuli could never be directly fixated but would always remain inside the scotoma. Although microsaccades were possible, these would not bring the visual stimulus into the seeing portion of the visual field. This methodology was the same as previous work (Fig 2B in [33]), in which we also provided examples of successfully identified saccades. Seven trials were removed from analysis in Experiment 1 and 2 trials from Experiment 2 due to eye movements of more than 1 degree towards the stimulus calculated from retrospective eye tracker data analysis. At the time of the experiment, a further 6 trials were flagged for exclusion in Experiment 1 and 4 in Experiment 2 due to real-time observation of the experimenter or feedback from the patient. In total, this accounted for 0.93% of trials in Experiment 1 and 0.54% of trials in Experiment 2 that were excluded from analysis due to inappropriate eye position.

fMRI procedure

The same stimuli were viewed during fMRI, presented separately to each hemifield. Stimuli during scanning were presented on a $1,280 \times 1,040$ resolution monitor at the back of the MRI scanner bore. Participants viewed stimuli via a double mirror mounted on the head coil. When in position, the screen subtended a visual angle of $23^{\circ} \times 13^{\circ}$. The same 5 speed levels were presented separately to each hemifield, representing a 10-condition block design (S2B Fig). For each block, the aperture of moving or stationary black dots appeared for 16 s. Direction coherence was 100%, and dots moved at a constant speed. Angle of drift changed at random every two seconds from a choice of 8 directions. A 10-s rest period followed each block. Throughout all experiments, participants performed a task to maintain fixation by pressing a button every time a central fixation cross changed colour from black to red (S2B Fig). Colour changes occurred at random, lasting 300 ms in duration, and participants were instructed at the start to try not to miss any red crosses. It was emphasised that they must try to maintain fixation throughout and avoid moving their eyes around the screen. An EyeLink 1000 eye tracker (SR Research Limited, Ontario, Canada) was again used to confirm central fixation by recording eye position (see section fMRI eye tracking).

MRI acquisition and preprocessing

Scanning took place using a 3T Siemens Verio MRI scanner at the Functional Magnetic Resonance Imaging Centre of the Brain (FMRIB, University of Oxford). At the start of each sequence, magnetisation was allowed to reach a steady state by discarding the first five volumes, an automated feature of the scanner. T2*-weighted EPI volumes covered 34 sequential 3-mm slices (repetition time, TR 2000 ms; echo time, TE 30 ms) with three runs, each lasting 260 s. In a single session lasting 13.2 min, 396 functional volumes were acquired. For one participant (P4) we collected one additional session of fMRI data. For three patients, we collected four runs of data (i.e., 526 volumes, P1, P2, P3). For one patient (P7) and one control, we only collected two runs of data, i.e., 266 volumes (8.9 min).

We also acquired a high-resolution (1 mm³) whole-head T1-weighted MPRAGE anatomical image (TE 4.68 ms; TR 2040 ms; flip angle, 8°) and a field map (TE1, 5.19 ms; TE2, 7.65 ms; 2 mm3) for each participant.

Quantification and statistical analysis

fMRI preprocessing and statistical analyses were carried out using tools from FMRIB's Software Library (FSL, www.fmrib.ox.ac.uk/fsl). Non-brain tissue was excluded from analysis using the Brain Extraction Tool (BET) [73], motion correction was carried out using MCFLIRT [74], and images were corrected for distortion using field maps. For cortical ROIs and group contrast maps, spatial smoothing used a Gaussian kernel of FWHM 5 mm, and high-pass temporal filtering (Gaussian-weighted least-squares straight line fitting, with sigma = 13.0 s) was applied. For all subcortical ROI analyses, no spatial smoothing was applied to ensure that signals were not contaminated with adjacent structures. Functional images were registered to high-resolution structural scans using FLIRT [75] and to a standard Montreal Neurological Institute (MNI) brain template using FLIRT and FNIRT [76]. This enabled us to transform anatomical and probabilistic regions of interest into functional space for analysis

(see <u>S3 Fig</u> for an illustration of subcortical ROIs in functional, structural, and standard space for each participant).

fMRI eye tracking

Eye movements during fMRI can be a legitimate concern when considering results for visual stimulation inside a scotoma. In this study, three main lines of evidence suggest that this was not a problem and could not have accounted for the results. First, concurrent eye movement data was collected on most patients using an eye tracker positioned at the base of the MRI bore (n = 10). All of these patients underwent successful eye-tracker calibration, with accurate data throughout fMRI runs. For this group of patients, the mean number of eye movements was 5.8 ± 3.5 SEM, defined as a movement of 1.5 degrees or more towards the scotoma. This accounted for <0.3% of the scan duration, suggesting that any effects on the results are likely to be negligible. To confirm this, when scanner volumes corresponding to eye movements were regressed out of analyses, the results remained unchanged (r > 0.99). For the patients without eye movement data, there had been difficulty either with calibration due to their dense field loss or with visualisation due to the presence of corrective acuity lenses. In those situations, direct visualisation was used via video recording of the pupil to observe any overt eye movements during the experiment.

Second, participants performed over 90% on a concurrent behavioural task that required fixation throughout the experiment (S2C Fig). Brief colour changes of the fixation cross (300-ms duration) occurred at frequent and random intervals, and participants were given a window of 1 s to press a button connected to the stimulus computer via a parallel port, being specifically instructed not to miss any red crosses or move their eyes around the screen. In addition, before the fMRI scan, all participants took part in behavioural testing lasting at least 60 min, focussed on their damaged region of vision. Participants became very experienced at maintaining fixation during this assessment.

Regions of interest

hMT+ masks were derived from probabilistic maps (Jülich atlas implemented in FSL) [77,78]. These were transformed into functional space for patients and controls to ensure consistency between participant groups. V1 masks in controls and in the undamaged hemisphere of patients were functionally defined so that they corresponded to stimulated regions of calcarine cortex. In native space, average hMT+ ROI volume was 94.8 ± 35.2 SD voxels in patients and 100.9 ± 42.0 SD voxels in controls (t = 0.5, p = 0.6, df = 42). Average V1 ROI volume was 16.2 ± 7.5 SD voxels in patients (undamaged hemisphere) and 24.4 ± 7.1 SD voxels in controls (averaged across hemispheres), the small volume reflective of the small 5°- or 8°-diameter stimulus used.

For the LGN and SC, binary masks were created by manual inspection and drawing over the anatomical T1-weighted images [79], using a radiological brain atlas to aid identification of landmarks (See S3 Fig for masks in all patients). The average LGN volume in patients measured 248 mm³ in the right and 246 mm³ in the left. In controls, average LGN volume was 240 mm³ in the right and 239 mm³ in the left. The average SC volume in patients was 195mm³ in the left and 177 mm³ in the right. For the ventral pulvinar, binary masks were created in MNI152 standard space according to the description of Arcaro and colleagues [50]. It was possible to visualise the nucleus as a region of low T1 intensity relative to surrounding tissue in the posterior most part of the thalamus. Masks were transformed to anatomical and functional space for each participant and were manually inspected to ensure accuracy (S3 Fig). Average ventral pulvinar volume was 385 mm³ in the left and 365 mm³ in the right. There were no significant differences in ROI volume between blindsight-positive and -negative patients.

Lesion masks

Lesion masks were drawn manually in structural space (S4 Fig). For S2D and S2E Fig, these were nonlinearly transformed to standard space and binarised before being summed. To assess whether lesions encroached upon hMT+ and/or its surrounding white matter, we created subject-specific cuboidal ROIs that were centred on the 'centre of gravity' of the ipsilesional hMT+ ROI in structural space. The isotropic cubes measured $40 \times 40 \times 40$ mm³, thus containing 64,000 1-mm³ voxels (see S5A and S5B Fig for examples in P6 and P12). We superimposed the binarized lesion mask over the cuboidal mask and counted the number of voxels that overlapped (red voxels in S5B Fig). The voxel counts are shown in the table in S5 Fig.

fMRI analysis

All graphs, signal change calculations, and correlation statistics were calculated using data from participants' native space.

For region of interest analysis, each experimental condition (e.g., left hemifield, 8°/s speed) was entered into the general linear model as a separate explanatory variable and was contrasted against the baseline fixation task to generate contrast of parameter estimates (COPEs) for each condition in every voxel. Signal change was then extracted from regions of interest within functional-space for each individual. The percentage of signal change was calculated by scaling the COPE by the peak-peak height of the regressor and dividing by the mean over time. These measures were averaged across participants to generate group plots for signal change as a function of the condition under investigation and were used in all correlation and regression analyses.

For whole-brain group analyses (Figs 4 and 5), it was necessary to align patient brains to a uniform pathological template with lesions located in the same 'left' hemisphere corresponding to a 'right-sided' visual field deficit. This required that the structural and functional images of three patients (P3, P5, P12) be flipped in the horizontal plane. All activation coordinates and images were in MNI space, with beta values displayed on mean structural images for the group transformed to standard space.

For the whole time series analyses, a value for residual BOLD signal in the ROI was obtained for each volume, and this was plotted against time. This was done separately for each participant and performed in functional space.

Brain imaging maps

As control participants demonstrated slightly different hMT+ localization in left and right hemispheres, we decided to show hMT+ group maps for left and right V1 lesions separately. This generated a group size of n = 6 for blindsight-positive patients with left V1 lesions (Fig 1Ai), n = 5 for blindsight-negative patients with left V1 lesions (Fig 1Bi), n = 2 for blindsight positive-patients with right V1 lesions (Fig 1Aii), and n = 1 for blindsight-negative patients with a right V1 lesion (Fig 1Bii). Mixed effects analyses were used for all group analyses where n > 3. A statistical threshold of p < 0.001 uncorrected was used to test for significance within V1 and extrastriate cortex, for which there were a priori hypotheses. Elsewhere, correction for multiple comparisons was made using a cluster threshold of p < 0.05 unless otherwise stated.

Quantitative statistics

Statistical tests to quantify differences in functional activity and co-activation between ROIs or participant groups were implemented in Excel or MATLAB. For overall hMT+ percent BOLD responses (Fig 1, bar charts), activity was averaged across all five motion speeds $(0-32^{\circ}/s)$. Two statistical analyses were then performed: (i) mean activity comparing sighted and blind hemi-field (and hemisphere) used a paired *t* test and (ii) mean activity compared to baseline using a one-sample *t* test versus zero. A two-way ANOVA was also used to assess the effect of participant group and speed on blind hemifield responses (right hemifield in controls) and separately on sighted hemifield responses (left hemifield in controls).

In fMRI time series correlations, it is known that task conditions can influence intrinsic temporal correlations (e.g., see [12,17]). To ensure that correlations only reflected resting block activity, we used the residuals timeseries for ROIs once stimulus responses had been regressed out. This allowed us to determine resting ROI1 versus ROI2 correlation analyses for each participant.

For correlation analyses, two main statistical methods were used. For correlations between participant groups, a Pearson correlation coefficient was derived from mean activity in ROI1 versus ROI2 at each level of speed, i.e., n = 5. For correlations between ROIs within participant groups, a Pearson correlation coefficient was determined separately for each participant (activity in ROI1 versus ROI2, at each level of speed). We then calculated weighted averages of r coefficients for the group, using a Fischer transformation to approximate correlations to a normally distributed measure (Figs 2 and 3). A significant effect of group was determined by performing a one-way ANOVA. Pairwise comparisons between participant groups or ROIs were then calculated using post-hoc *t* tests. One-sample *t* tests were used to compare r coefficients to zero. Whenever activity was compared in the same participant, a paired correlation analysis was performed.

Seed region correlation maps

For each participant, the raw signal time series for the seed ROI was entered into the model as an explanatory variable. This was applied to the filtered and motion corrected whole brain timecourse. Stimulus conditions were also entered as regressors so that the model would better describe the data. Since the model was identical to the ROI time series, the parameter estimate was always 1 in the seed region. Any participant with voxels outside the seed region with a parameter estimate >1.5 were excluded from analysis, as these results were likely to be driven by noise. This was only a problem when using subcortical structures as seeds, as these are small regions with relatively weak signal that are more susceptible to artefact. This led to the exclusion of three participants from subcortical seed analyses (both hemispheres); two were blindsight positive (P5, P10), and one was blindsight negative (P7). Corroborating this, the raw LGN signal range in those three participants was significantly greater than the other 11 patients ($66.6 \pm$ SD 10.5 versus $39.6 \pm$ SD 6.07, t = 2.2, p = 0.04) and controls (30.6 ± 1.9 , t = 5.2, p < 0.001, df = 12).

To generate seed region correlation maps (Figs 4 and 5), the COPEs for each included participant were entered into a higher-level mixed effects analysis, and output parameter estimate (beta) maps were used to represent seed region correlation maps. This was performed separately for each participant group and for each seed region. The resulting maps were not intended to determine statistical significance but to allow visual inspection of the results from the separate groups. Visualisation thresholds were based upon control participants, and an optimal cutoff was used to display correlations in either V1, visual subcortex, and/or hMT + without excess background noise. The same thresholds were applied to all participant groups. For LGN-oriented maps (Fig 5Aiv–5Avi and 5Biv–5Bvi), although not shown, no other subcortical regions showed equivalent beta levels, although ventral pulvinar was co-activated in both hemispheres of all participant groups when using a reduced threshold of 0.32.

Supporting information

S1 Table. Pathology location and patient demographics. Descriptions include pathology nature and anatomical location, gender, age at participation in the study, time since pathology onset (months), visual field deficit, and blindsight status. HH = homonymous hemianopia, L = left, M = months, R = right, UQ = upper quadrantanopia. (DOCX)

S1 Fig. Behavioural paradigm and results. (A) Experiment 1: 2AFC temporal detection. Patients fixated on a central cross, with onset of each 1,500-ms interval alerted by a low (interval 1) or high (interval 2) pitch tone. Stimuli were located inside the scotoma (see S6 Fig) and could appear in either interval at random for a period of 500 ms. At the end of the trial, participants had to decide in which interval it appeared. Stimuli consisted of an aperture of 5° or 8° diameter, containing static or moving black dots (speed 0, 4, 8, 20, or 32°/s, at random). (B) Experiment 2: 2AFC direction discrimination. Throughout each trial of 2,500-ms duration, participants were required to fixate on a central black cross. During this time, the stimulus appeared inside the scotoma for 500 ms with jittered onset. At the end of the trial, patients had to indicate which direction the dots were moving (horizontal or vertical). If they saw nothing, they were instructed to guess. Controls did not perform behavioural experiments, as they would be at ceiling. (C) Mean behavioural performance \pm SEM for 2AFC temporal detection, as a function of stimulus speed. (D) Mean behavioural performance \pm SEM for 2AFC direction discrimination, as a function of stimulus speed. Results for blindsight-positive patients (pink diamond) and blindsight-negative patients (blue circle) are shown separately. Dashed grey line represents chance level (50%), with statistical symbols representing group-level one-tailed ttests versus chance (μ : p < 0.01, *: p < 0.05). All other values were nonsignificant. Underlying data can be found in S4 Data. 2AFC, two-alternate forced choice; SEM, standard error of the mean.



S2 Fig. fMRI procedure and group lesion maps. (A) Stimulus size and position for all patients and controls. Each transparent circle represents the stimulus aperture for a single participant. The black cross at coordinates (0,0) represents fixation. Only the right hemifield is shown, but stimuli were also presented to precisely equivalent locations in the opposite hemifield. (B) Simple block design, presenting an aperture of black dots to the blind portion of visual field or its equivalent location in the sighted hemifield. Stimuli had identical parameters to behavioural testing. Stimulus speed in each block was randomized to one of five levels (0, 4, 8, 20, and 32°/s). This represented 10 conditions in total, with each block lasting 16 s with 10-s rest periods. (C) Throughout all blocks, a fixation task required participants to press a button every time the central fixation cross changed colour from black to red. Colour changes occurred at random lasting 300 ms. All participants scored at least 90%, with mean performance and SEM plotted for patients and controls. (D) Summed lesion maps for blindsightpositive patients on standard-space MNI template brain. (E) Summed lesion maps for blindsight-negative patients, on standard space MNI template brain. Colour scale represents the number of patients with lesions involving that voxel, from 1 to 8 in blindsight-positive patients and 1 to 6 in blindsight-negative patients. Patients with right hemisphere lesions (n = 3) had structural scans flipped in the horizontal plane to allow images to be aligned and aid

visualisation, using radiological convention. Shaded blue areas represent binarized Jülichdefined probabilistic maps of hMT+. Underlying data for panels A and C can be found in <u>\$5</u> Data. fMRI, functional MRI; MNI, Montreal Neurological Institute; SEM, standard error of the mean.

(TIF)

S3 Fig. Subcortical ROIs in native (functional), structural, and standard space for all 14 patients. Slices in each 'space' are matched to show equivalent, representative views. Ventral pulvinar (yellow) and SC (blue) ROIs are depicted in the upper rows and LGN (green) in lower rows. Background brain images in structural space are the T1-weighted MPRAGE images. These have been transformed to native (functional) and standard space for background images, radiological convention. LHF refers to blind left hemifield, RHF refers to blind right hemifield, BS+ is blindsight positive, and BS- is blindsight negative. LGN, lateral geniculate nucleus; ROI, region of interest; SC, superior colliculus. (PNG)

S4 Fig. Individual lesion maps. (A) Blindsight-positive patients. **(B)** Blindsight-negative patients. Lesions are highlighted with red-yellow masks, and shaded blue areas represent binarized Jülich-defined probabilistic maps of hMT+. Background images are structural T1 scans, radiological convention. (PNG)

S5 Fig. Lesion involvement of hMT+ and its surrounding white matter. (A) Example of P6 and (B) P12, illustrating the isotropic $40 \times 40 \times 40$ mm³ cuboidal ROI (green), centred on the ipsilesional hMT+ mask (blue), which includes hMT+ surrounding white matter. Where the ROI overlaps with the lesion mask, voxels are coloured red (no overlap in panel A). Back-ground brain images are structural T1 scans, radiological convention. (C) The voxel count for regions of overlap between the hMT+/white matter ROI and lesion masks for each patient. ROI, region of interest.

(PNG)

S6 Fig. Visual field loss and T1 structural images. In each patient, perimetry reports are depicted schematically showing the location of target stimuli. Dense visual field loss is shown in black (<0.5%) and partial loss in grey (<2%). Stimuli were restricted to a region of dense visual field loss, a minimum of 2.5 degrees from fixation. Concentric rings represent increments in retinal position of 10 degrees, spanning the central 30 degrees. Equivalent perimetry data (Humphrey 30:2 except P10, who has Goldmann) are shown alongside (outer columns) where available. Blindsight status and estimates of the percentage of scotoma covered by the stimulus (percent) are provided for each patient. Representative T1 structural axial slices demonstrate the lesion location, using radiological convention. Of note, patients P3, P8, P10, P11, and P13 took part in a previous study [33], in which they demonstrated significant blindsight performance for detection of a drifting Gabor. (PNG)

S1 Data. Mean signal change (%) in contralateral hMT+, (1) averaged across all 5 speed conditions, (2) as a function of stimulus speed, for each hemifield in patients and controls. This data underlies Fig 1iii and 1iv. (XLSX)

S2 Data. Residual BOLD signal in the ROIs, for every scan volume. Each tab represents data for one participant. This data underlies Fig 2. BOLD, blood oxygen level-dependent; ROI,

region of interest. (XLSX)

S3 Data. Residual BOLD signal in the ROIs, for every scan volume. Each tab represents data for one participant. This data underlies Fig 3. BOLD, blood oxygen level–dependent; ROI, region of interest. (XLSX)

S4 Data. Mean behavioural performance for 2AFC detection and discrimination tests, for each participant. This data underlies <u>S1 Fig.</u> 2AFC, two-alternate forced choice. (XLSX)

S5 Data. (1) Position and size of stimulus apertures, and (2) performance on the fMRI fixation task, for each participant. This data underlies <u>S2A and S2C Fig</u>. fMRI, functional MRI. (XLSX)

Acknowledgments

The authors wish to thank all the participants for taking part in this study.

Author Contributions

Conceptualization: Sara Ajina, Holly Bridge.

Data curation: Sara Ajina.

Formal analysis: Sara Ajina.

Funding acquisition: Sara Ajina, Holly Bridge.

Investigation: Sara Ajina.

Methodology: Sara Ajina, Holly Bridge.

Project administration: Sara Ajina.

Supervision: Holly Bridge.

Writing - original draft: Sara Ajina.

Writing - review & editing: Sara Ajina, Holly Bridge.

References

- Zhang X, Kedar S, Lynn MJ, Newman NJ, Biousse V. Homonymous hemianopias. Clinical–anatomic correlations in 904 cases. Neurology. 2006; 66: 906–910. https://doi.org/10.1212/01.wnl.0000203913. 12088.93 PMID: 16567710
- Weiskrantz L, Warrington EK, Sanders MD, Marshall J. Visual capacity in the hemianopic field following a restricted occipital ablation. Brain. 1974; 97: 709–28. PMID: 4434190
- 3. Ajina S, Pestilli F, Rokem A, Kennard C, Bridge H. Human blindsight is mediated by an intact geniculoextrastriate pathway. eLife. 2015; 4, e08935. https://doi.org/10.7554/eLife.08935 PMID: 26485034
- 4. Le Bihan D, Johansen-Berg H. Diffusion MRI at 25: Exploring brain tissue structure and function. Neuroimage. 2012; 61(2): 324–341. https://doi.org/10.1016/j.neuroimage.2011.11.006 PMID: 22120012
- Leh SE, Johansen-Berg H, Ptito A. Unconscious vision: new insights into the neuronal correlate of blindsight using diffusion tractography. Brain. 2006; 129, 1822–1832. https://doi.org/10.1093/brain/awl111 PMID: 16714319
- Bridge H, Thomas O, Jbabdi S, Cowey A. Changes in connectivity after visual cortical brain damage underlie altered visual function. Brain. 2008; 131: 1433–1444. https://doi.org/10.1093/brain/awn063 PMID: 18469021

- Bridge H, Hicks SL, Xie J, Okell TW, Mannan S, Alexander I, Cowey A, Kennard C. Visual activation of extra-striate cortex in the absence of V1 activation. Neuropsychologia. 2010; 48:4148–4154. https:// doi.org/10.1016/j.neuropsychologia.2010.10.022 PMID: 20974160
- Warner CE, Kwan WC, Wright D, Johnston LA, Egan GF, Bourne JA. Preservation of vision by the pulvinar following early-life primary visual cortex lesions. Curr Biol. 2015; 25(4):424–34. https://doi.org/10. 1016/j.cub.2014.12.028 PMID: 25601551
- Mundinano IC, Chen J, de Souza M, Sarossy MG, Joanisse MF, Goodale MA, et al. More than blindsight: Case report of a child with extraordinary visual capacity following perinatal bilateral occipital lobe injury. Neuropsychologia. 2017; pii: S0028-3932(17)30430-X.
- Bola M, Gall C, Sabel BA. The second face of blindness: processing speed deficits in the intact visual field after pre- and post-chiasmatic lesions. PLoS One. 2013; 8(5):e63700 https://doi.org/10.1371/ journal.pone.0063700 PMID: 23667657
- Perez C, Peyrin C, Cavézian C, Coubard O, Caetta F, Raz N, et al. An FMRI investigation of the cortical network underlying detection and categorization abilities in hemianopic patients. Brain Topogr. 2013; 26 (2):264–77. https://doi.org/10.1007/s10548-012-0244-z PMID: 22878845
- Rogers BP, Morgan VL, Newton AT, Gore JC. Assessing functional connectivity in the human brain by fMRI. Magn Reson Imaging. 2007; 25(10):1347–57. <u>https://doi.org/10.1016/j.mri.2007.03.007</u> PMID: 17499467
- Smith SM, Vidaurre D, Beckmann CF, Glasser MF, Jenkinson M, Miller KL, et al. Functional connectomics from resting-state fMRI. Trends Cogn Sci. 2013; 17(12), 666–82. https://doi.org/10.1016/j.tics. 2013.09.016 PMID: 24238796
- Benevento LA, Standage GP. The organization of projections of the retinorecipient and nonretinorecipient nuclei of the pretectal complex and layers of the superior colliculus to the lateral pulvinar and medial pulvinar in the macaque monkey. Journal of Comparative Neurology. 1983; 217(3):307–336. https://doi.org/10.1002/cne.902170307 PMID: 6886056
- Leh SE, Chakravarty MM, Ptito A. The connectivity of the human pulvinar: a diffusion tensor imaging tractography study. Int J Biomed Imaging. 2008; 789539.
- Purushothaman G, Marion R, Li K, Casagrande VA. Gating and control of primary visual cortex by pulvinar. Nat Neurosci. 2012; 15(6):905–12. https://doi.org/10.1038/nn.3106 PMID: 22561455
- Hampson M, Olson IR, Leung HC, Skudlarski P, Gore JC. Changes in functional connectivity of human MT/V5 with visual motion input. Neuroreport. 2004; 15(8):1315–9. PMID: 15167557
- Schmid MC, Panagiotaropoulos T, Augath MA, Logothetis NK, Smirnakis SM. Visually driven activation in macaque areas V2 and V3 without input from the primary visual cortex. PLoS One. 2009; 4(5): e5527. https://doi.org/10.1371/journal.pone.0005527 PMID: 19436733
- Margulies DS, Vincent JL, Kelly C, Lohmann G, Uddin LQ, Biswal BB, et al. Precuneus shares intrinsic functional architecture in humans and monkeys. Proc Natl Acad Sci U S A. 2009; 106(47):20069–74. https://doi.org/10.1073/pnas.0905314106 PMID: 19903877
- Honey CJ, Sporns O, Cammoun L, Gigandet X, Thiran JP, Meuli R, et al. Predicting human restingstate functional connectivity from structural connectivity. Proc Natl Acad Sci U S A. 2009; 106(6):2035– 40. https://doi.org/10.1073/pnas.0811168106 PMID: 19188601
- Huntenburg JM, Bazin PL, Goulas A, Tardif CL, Villringer A, Margulies DS. A Systematic Relationship Between Functional Connectivity and Intracortical Myelin in the Human Cerebral Cortex. Cereb Cortex. 2017; 27(2):981–997. https://doi.org/10.1093/cercor/bhx030 PMID: 28184415
- Van Dijk KR, Hedden T, Venkataraman A, Evans KC, Lazar SW, Buckner RL. Intrinsic functional connectivity as a tool for human connectomics: theory, properties, and optimization. J Neurophysiol. 2010; 103(1):297–321. https://doi.org/10.1152/jn.00783.2009 PMID: 19889849
- 23. Biswal B, Yetkin FZ, Haughton VM, Hyde JS. Functional connectivity in the motor cortex of resting human brain using echo-planar MRI. Magn Reson Med. 1995; 34(4):537–41. PMID: 8524021
- Quigley M, Cordes D, Wendt G, Turski P, Moritz C, Haughton V, et al. Effect of focal and nonfocal cerebral lesions on functional connectivity studied with MR imaging. AJNR Am J Neuroradiol. 2001; 22 (2):294–300. PMID: 11156772
- Koch MA, Norris DG, Hund-Georgiadis M. An investigation of functional and anatomical connectivity using magnetic resonance imaging. Neuroimage. 2002; 16(1):241–50. https://doi.org/10.1006/nimg. 2001.1052 PMID: 11969331
- Lyon DC, Nassi JJ, Callaway EM. A disynaptic relay from superior colliculus to dorsal stream visual cortex in macaque monkey. Neuron. 2010; 65(2):270–9. https://doi.org/10.1016/j.neuron.2010.01.003 PMID: 20152132
- Barbur JL, Watson JD, Frackowiak RS, Zeki S. Conscious visual perception without V1.Brain. 1993; 116 (6):1293–302

- Holliday IE, Anderson SJ, Harding GF. Magnetoencephalographic evidence for non-geniculostriate visual input to human cortical area V5. Neuropsychologia. 1997; 35(8):1139–46. PMID: 9256379
- Zeki S, Ffytche DH. The Riddoch syndrome: insights into the neurobiology of conscious vision.Brain. 1998; 121 (1):25–45.
- Goebel R, Muckli L, Zanella FE, Singer W, Stoerig P. Sustained extrastriate cortical activation without visual awareness revealed by fMRI studies of hemianopic patients. Vision Res. 2001; 41(10–11):1459– 74. PMID: 11322986
- Schoenfeld MA, Noesselt T, Poggel D, Tempelmann C, Hopf JM, Woldorff MG, et al. Analysis of pathways mediating preserved vision after striate cortex lesions. Ann Neurol. 2002; 52(6):814–24. https://doi.org/10.1002/ana.10394 PMID: 12447936
- Ajina S, Kennard C, Rees G, Bridge H. Motion area V5/MT+ response to global motion in the absence of V1 resembles early visual cortex. Brain. 2015; 138(1), 164–78.
- Ajina S, Rees G, Kennard C, Bridge H. Abnormal contrast responses in the extrastriate cortex of blindsight patients. J. Neurosci. 2015; 35(21), 8201–8213. https://doi.org/10.1523/JNEUROSCI.3075-14. 2015 PMID: 26019336
- 34. Gaglianese A, Costagli M, Ueno K, Ricciardi E, Bernardi G, Pietrini P, et al. The direct, not V1-mediated, functional influence between the thalamus and middle temporal complex in the human brain is modulated by the speed of visual motion. Neuroscience. 2015; 22; 284: 833–44. <u>https://doi.org/10.1016/j.neuroscience.2014.10.042</u> PMID: 25450965
- Rodman HR, Gross CG, Albright TD. Afferent basis of visual response properties in area MT of the macaque. I. Effects of striate cortex removal. J Neurosci. 1989; 9(6): 2033–50. PMID: 2723765
- Azzopardi P, Fallah M, Gross CG, Rodman HR. Response latencies of neurons in visual areas MT and MST of monkeys with striate cortex lesions. Neuropsychologia. 2003; 41(13):1738–56. PMID: 14527538
- Schmid MC, Mrowka SW, Turchi J, Saunders RC, Wilke M, Peters AJ, et al. Blindsight depends on the lateral geniculate nucleus. Nature. 2010; 466, 373–377. <u>https://doi.org/10.1038/nature09179</u> PMID: 20574422
- ffytche DH, Guy CN, Zeki S. The parallel visual motion inputs into areas V1 and V5 of human cerebral cortex. Brain. 1995; 118 (6):1375–94.
- Benevento LA, Yoshida K. The afferent and efferent organization of the lateral geniculo-prestriate pathways in the macaque monkey. J Comp Neurol. 1981; 203(3):455–74. <u>https://doi.org/10.1002/cne.</u> 902030309 PMID: 6274921
- Cowey A, Stoerig P. Projection patterns of surviving neurons in the dorsal lateral geniculate nucleus following discrete lesions of striate cortex: implications for residual vision. Exp Brain Res. 1989; 75 (3):631–8. PMID: 2744120
- Fries W. The projection from the lateral geniculate nucleus to the prestriate cortex of the macaque monkey. Proc R Soc Lond B Biol Sci. 1981; 213(1190):73–86. PMID: 6117869
- Sincich LC, Park KF, Wohlgemuth MJ, Horton JC. Bypassing V1: a direct geniculate input to area MT. Nat Neurosci. 2004; 7(10):1123–8. https://doi.org/10.1038/nn1318 PMID: 15378066
- Stepniewska I, Qi HX, Kaas JH. Do superior colliculus projection zones in the inferior pulvinar project to MT in primates? Eur J Neurosci. 1999; 11(2):469–80. PMID: 10051748
- Chen W, Zhu XH, Thulborn KR, Ugurbil K. Retinotopic mapping of lateral geniculate nucleus in humans using functional magnetic resonance imaging. Proc. Natl. Acad. Sci. USA. 1999; 96(5), 2430–2434. PMID: 10051659
- Burgel U, Amunts K, Hoemke L, Mohlberg H, Gilsbach JM, Zilles K. et al. White matter fiber tracts of the human brain: three-dimensional mapping at microscopic resolution, topography and intersubject variability. Neuroimage. 2006; 29, 1092–1105. <u>https://doi.org/10.1016/j.neuroimage.2005.08.040</u> PMID: 16236527
- 46. Sillito AM, Cudeiro J, Jones HE. Always returning: feedback and sensory processing in visual cortex and thalamus. Trends Neurosci. 2006; 29(6):307–16. https://doi.org/10.1016/j.tins.2006.05.001 PMID: 16713635
- Ajina S, Bridge H. Subcortical pathways to extrastriate visual cortex underlie residual vision following bilateral damage to V1. Neuropsychologia. 2018; pii: S0028-3932(18)30006-X.
- Berman RA, Wurtz RH. Functional identification of a pulvinar path from superior colliculus to cortical area MT.J Neurosci. 2010; 30(18):6342–54. https://doi.org/10.1523/JNEUROSCI.6176-09.2010 PMID: 20445060
- **49.** Warner CE, Goldshmit Y, Bourne JA. Retinal afferents synapse with relay cells targeting the middle temporal area in the pulvinar and lateral geniculate nuclei. Front Neuroanat. 2010; 4(8):1–16.

- Arcaro MJ, Pinsk MA, Kastner S. The Anatomical and Functional Organization of the Human Visual Pulvinar. J. Neurosci. 2015; 35(27), 9848–9871. <u>https://doi.org/10.1523/JNEUROSCI.1575-14.2015</u> PMID: 26156987
- Priebe NJ, Cassanello CR, Lisberger SG. The neural representation of speed in macaque area MT/V5. J Neurosci. 2003; 23(13):5650–61. PMID: 12843268
- Nover H, Anderson CH, DeAngelis GC. A logarithmic, scale-invariant representation of speed in macaque middle temporal area accounts for speed discrimination performance. J Neurosci. 2005; 25 (43):10049–60. https://doi.org/10.1523/JNEUROSCI.1661-05.2005 PMID: 16251454
- Chawla D, Buechel C, Edwards R, Howseman A, Josephs O, Ashburner J, et al. Speed-dependent responses in V5: A replication study. Neuroimage. 1999; 9(5), 508–515. https://doi.org/10.1006/nimg. 1999.0432 PMID: 10329290
- 54. Bender DB. Receptive-field properties of neurons in the macaque inferior pulvinar. J Neurophysiol. 1982; 48(1):1–17. https://doi.org/10.1152/jn.1982.48.1.1 PMID: 7119838
- Casanova C, Savard T. Responses to moving texture patterns of cells in the striate-recipient zone of the cat's lateral posterior-pulvinar complex. Neuroscience. 1996; 70(2):439–47. PMID: 8848152
- Bender DB. Visual activation of neurons in the primate pulvinar depends on cortex but not colliculus. Brain Res. 1983; 279(1–2):258–61. PMID: <u>6640346</u>
- 57. Lee BB, Creutzfeldt OD, Elepfandt A. The responses of magno- and parvocellular cells of the monkey's lateral geniculate body to moving stimuli. Exp Brain Res. 1979; 35(3):547–57. PMID: <u>110614</u>
- Sahraie A, Weiskrantz L, Barbur JL, Simmons A, Williams SC, Brammer MJ. Pattern of neuronal activity associated with conscious and unconscious processing of visual signals. Proc Natl Acad Sci. 1997; 94 (17):9406–11. PMID: 9256495
- Tamietto M, Cauda F, Corazzini LL, Savazzi S, Marzi CA, Goebel R, et al. Collicular vision guides nonconscious behavior. J Cogn Neurosci.2 2010; 2(5):888–902.
- Rodman HR, Gross CG, Albright TD. Afferent basis of visual response properties in area MT of the macaque. II. Effects of superior colliculus removal. J Neurosci. 1990; 10(4): 1154–64. PMID: 2329373
- Mohler CW, Wurtz RH. Role of striate cortex and superior colliculus in visual guidance of saccadic eye movements in monkeys. J Neurophysiol. 1977; 40(1): 74–94. https://doi.org/10.1152/jn.1977.40.1.74 PMID: 401874
- Kato R, Takaura K, Ikeda T, Yoshida M, Isa T. (2011). Contribution of the retino-tectal pathway to visually guided saccades after lesion of the primary visual cortex in monkeys. Eur J Neurosci. 33(11): 1952–60. https://doi.org/10.1111/j.1460-9568.2011.07729.x PMID: 21645091
- 63. Azzopardi P, Cowey A. Motion discrimination in cortically blind patients. Brain. 2001; 124(1), 30–46.
- Chawla D, Phillips J, Buechel C, Edwards R, Friston KJ. Speed-dependent motion-sensitive responses in V5: an fMRI study. Neuroimage. 1998; 7(2), 86–96. https://doi.org/10.1006/nimg.1997.0319 PMID: 9558641
- Martin T, Das A, Huxlin KR. Visual cortical activity reflects faster accumulation of information from cortically blind fields. Brain. 2012; 135(11):3440–52.
- Ajina S, Bridge H. Blindsight and Unconscious Vision: What They Teach Us about the Human Visual System. Neuroscientist. 2016; pii: 1073858416673817.
- 67. Brainard DH. The Psychophysics Toolbox. Spatial Vision 1997; 10, 433–436. PMID: 9176952
- 68. Kleiner M, Brainard DH, Pelli D. What's new in Psychtoolbox-3? Perception 2007; 36, ECVP Abstract Supplement.
- 69. Morland AB, Lê S, Carroll E, Hoffmann MB, Pambakian A. The role of spared calcarine cortex and lateral occipital cortex in the responses of human hemianopes to visual motion. J Cogn Neurosci. 2004; 16(2): 204–18. https://doi.org/10.1162/089892904322984517 PMID: 15068592
- 70. Weiskrantz L, Barbur JL, Sahraie A. Parameters affecting conscious versus unconscious visual discrimination with damage to the visual cortex (V1). Proc Natl Acad Sci U S A. 1995; 92(13): 6122–6. PMID: 7597090
- Zihl J, von Cramon D. Registration of light stimuli in the cortically blind hemifield and its effect on localization. Behav Brain Res. 1980; 1(4):287–98. PMID: 7295378
- 72. de Gelder B, Tamietto M, van Boxtel G, Goebel R, Sahraie A, van den Stock J, et al. Intact navigation skills after bilateral loss of striate cortex. Curr Biol. 2008; 18(24): R1128–9. <u>https://doi.org/10.1016/j.cub.2008.11.002</u> PMID: 19108766
- 73. Smith SM. Fast robust automated brain extraction. Hum. Brain Map. 2002; 17(3), 143–155.
- 74. Jenkinson M, Bannister P, Brady JM, Smith SM. Improved Optimisation for the Robust and Accurate Linear Registration and Motion Correction of Brain Images. NeuroImage. 2002; 17(2), 825–841. PMID: 12377157

- Jenkinson M, Smith SM. A global optimisation method for robust affine registration of brain images. Medical Image Analysis. 2001; 5(2), 143–156. PMID: <u>11516708</u>
- Andersson JL, Jenkinson M, Smith SM. Non-linear optimisation. FMRIB technical report; 2007; TR07JA1.
- 77. Malikovic A, Amunts K, Schleicher A, Mohlberg H, Eickhoff SB, Wilms M, et al. Cytoarchitectonic analysis of the human extrastriate cortex in the region of V5/MT+: a probabilistic, stereotaxic map of area hOc5. Cereb. Cortex. 2007; 17(3), 562–574. https://doi.org/10.1093/cercor/bhj181 PMID: 16603710
- Eickhoff SB, Stephan KE, Mohlberg H, Grefkes C, Fink GR, Amunts K, et al. A new SPM toolbox for combining probabilistic cytoarchitectonic maps and functional imaging data. Neuroimage. 2005; 25 (4):1325–35. https://doi.org/10.1016/j.neuroimage.2004.12.034 PMID: 15850749
- 79. Horton JC, Landau K, Maeder P, Hoyt WF. Magnetic Resonance Imaging of the Human Lateral Geniculate Body. Arch. Neurol. 1990; 47(11),1201–1206. PMID: 2241617